Experimental Infection of the Pig with *Mycobacterium ulcerans*: A Novel Model for Studying the Pathogenesis of Buruli Ulcer Disease



Miriam Bolz^{1,2}, Nicolas Ruggli³, Marie-Thérèse Ruf^{1,2}, Meret E. Ricklin³, Gert Zimmer³, Gerd Pluschke^{1,2}*

1 Swiss Tropical and Public Health Institute, Basel, Switzerland, 2 University of Basel, Basel, Switzerland, 3 Institute of Virology and Immunology (IVI), Mittelhäusern, Switzerland

Abstract

Background: Buruli ulcer (BU) is a slowly progressing, necrotising disease of the skin caused by infection with *Mycobacterium ulcerans*. Non-ulcerative manifestations are nodules, plaques and oedema, which may progress to ulceration of large parts of the skin. Histopathologically, BU is characterized by coagulative necrosis, fat cell ghosts, epidermal hyperplasia, clusters of extracellular acid fast bacilli (AFB) in the subcutaneous tissue and lack of major inflammatory infiltration. The mode of transmission of BU is not clear and there is only limited information on the early pathogenesis of the disease available.

Methodology/Principal Findings: For evaluating the potential of the pig as experimental infection model for BU, we infected pigs subcutaneously with different doses of *M. ulcerans*. The infected skin sites were excised 2.5 or 6.5 weeks after infection and processed for histopathological analysis. With doses of 2×10^7 and 2×10^6 colony forming units (CFU) we observed the development of nodular lesions that subsequently progressed to ulcerative or plaque-like lesions. At lower inoculation doses signs of infection found after 2.5 weeks had spontaneously resolved at 6.5 weeks. The observed macroscopic and histopathological changes closely resembled those found in *M. ulcerans* disease in humans.

Conclusion/Significance: Our results demonstrate that the pig can be infected with *M. ulcerans*. Productive infection leads to the development of lesions that closely resemble human BU lesions. The pig infection model therefore has great potential for studying the early pathogenesis of BU and for the development of new therapeutic and prophylactic interventions.

Citation: Bolz M, Ruggli N, Ruf M-T, Ricklin ME, Zimmer G, et al. (2014) Experimental Infection of the Pig with *Mycobacterium ulcerans*: A Novel Model for Studying the Pathogenesis of Buruli Ulcer Disease. PLoS Negl Trop Dis 8(7): e2968. doi:10.1371/journal.pntd.0002968

Editor: Pamela L. C. Small, University of Tennessee, United States of America

Received April 22, 2014; Accepted May 31, 2014; Published July 10, 2014

Copyright: © 2014 Bolz et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability: The authors confirm that all data underlying the findings are fully available without restriction. All relevant data are within the paper and its Supporting Information files.

Funding: This work was funded in part by the Stop Buruli Initiative supported by the UBS-Optimus Foundation. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* Email: gerd.pluschke@unibas.ch

Introduction

Buruli ulcer (BU), caused by infection with *Mycobacterium ulcerans*, is a human disease of the skin primarily affecting subcutaneous fat tissue and leading to ulceration of the overlying dermal and epidermal layers [1,2]. The disease is reported from countries worldwide but has its highest prevalence in West Africa [3]. Natural reservoirs of *M. ulcerans* as well as the mode(s) of transmission are not clearly identified [3,4]. While for a long time wide surgical excision was the only treatment option for BU, since 2004 the World Health Organization (WHO) recommends antibiotic therapy with rifampicin and streptomycin for 8 weeks [5]. This change in standard treatment has reduced recurrence rates to less than 2% [6–9].

M. ulcerans produces the polyketide exotoxin mycolactone that is responsible for the necrotizing nature of BU [10]. Three distinct non-ulcerative stages of the disease are described: subcutaneous, painless and movable nodules or papules, oedema and plaques. All

three stages may progress to ulceration once the destruction of the subcutis results in collapse of the overlying epidermis and dermis [11].

Ulcerative BU lesions have been histopathologically well described through the analysis of excised tissue from surgically treated patients. Coagulative necrosis, fat cell ghosts and epidermal hyperplasia together with the presence of extracellular clusters of acid fast bacilli (AFB) in the absence of major inflammatory infiltrates in central parts of the lesions are considered hallmarks of the disease and can also be used for histopathological diagnosis [12,13]. However, early, preulcerative stages have been described less frequently, because in particular in the African BU endemic regions patients are rarely reporting at treatment centres during early stages of the disease. Furthermore, with the replacement of surgical treatment by chemotherapy, tissue samples are not easily available any longer. Therefore, a suitable experimental animal infection model is required to contribute to the understanding of early host-pathogen interactions and pathogenesis in BU.

Author Summary

Buruli ulcer caused by Mycobacterium ulcerans infection is a necrotizing disease of the skin and the underlying subcutaneous tissue. Since the skin of pigs (Sus scrofa) has striking structural and physiological similarities with human skin, we investigated whether it is possible to develop an experimental M. ulcerans infection model by subcutaneous injection of the mycobacteria into pig skin. Injection of 2×10^6 or 2×10^7 colony forming units of M. ulcerans led to the development of lesions that were both macroscopically and microscopically very similar to human Buruli ulcer lesions. In particular for the characterization of the pathogenesis of Buruli ulcer and of immune defence mechanisms against *M. ulcerans*, the pig model appears to be superior to the mouse foot pad model commonly used for the evaluation of the efficacy of chemotherapeutic regimens.

A range of animal species have been reported of being naturally infected with *M. ulcerans* and of developing ulcerative lesions. These include koalas, possums, cats, dogs and horses [14–21]. Except for possums which appear to be unusually susceptible to the disease, these animal infections seem to occur only sporadically [22]. Experimental *M. ulcerans* infections have been performed with amphibians, armadillos, rats, mice, guinea pigs and monkeys, with a mouse foot pad model being most widely used for studying the efficacy of prophylactic and therapeutic interventions [23–29]. Here we propose the pig (*Sus scrofa*) as experimental *M. ulcerans* infection model, since pigs are closely related to humans in terms of many aspects of anatomy and physiology [30,31]. The pig is widely used as a model in dermatological studies because pig skin, in contrast to rodent skin, has striking similarities to human skin [32]. Not only the thickness of the epidermis and the dermis are comparable to human skin [33], but also the presence of a subcutaneous fat cell layer is favouring the pig model over the mouse foot pad model commonly used for analysing BU pathogenesis. Furthermore, the porcine immune system reflects the human immune system in many aspects better than the murine immune system does [34,35]. For all these reasons we explored here the potential of the pig to serve as model for human M. *ulcerans* infection.

Materials and Methods

Ethical statement

All animal experiments described here were approved by the Animal Welfare Committee of the Canton of Berne under licence number BE50/11, and conducted in compliance with the Swiss animal protection law and with other national and international guidelines.

Bacteria

The *M. ulcerans* strain used in this study was isolated in 2010 from a swab taken from the undermined edges of the ulcerative lesion of a Cameroonian BU patient [4]. Five passages of the strain after isolation were done in Bac/T medium (Biomerieux) at 30°C. For preparation of the inoculum, bacteria were cultivated in Bac/T medium for 6 weeks, recovered by centrifugation and diluted in sterile phosphate-buffered saline (PBS) to 375 mg/ml wet weight corresponding to 2×10^8 CFU/ml as determined by plating serial dilutions on 7H9 agar plates. From this stock solution suspension serial dilutions in PBS were prepared for infection.



Figure 1. Macroscopic appearance of pig skin infected with *M. ulcerans.* Development of representative lesions 2.5 weeks or 6.5 weeks after subcutaneous infection with 2×10^7 (B1, D1 and E1) or 2×10^6 (C1) CFU is depicted and compared to a site left uninfected (A1). Excised tissue specimens were fixed and vertically cut in half to visualize macroscopically visible alteration in tissue structure (A2–E2). For high inoculation doses ($\geq 2 \times 10^6$ CFU) the formation of nodules (B1) with necrotic centres was observed already 2.5 weeks after infection (B2). These yellow centres indicative for coagulative necrosis were surrounded by a reddish ring (B2). At 6.5 weeks after infection, these nodules had progressed into a small ulcer or a plaque (D1, E1) associated with marked macroscopically visible alterations in tissue structure (D2, E2). At sites injected with 2×10^6 CFU nodules with greyish discoloration of the dermis had developed 6.5 weeks after injection of *M. ulcerans* (C1, C2). doi:10.1371/journal.pntd.0002968.q001

Figure 2. Microscopic appearance of pig skin 2.5 weeks after experimental infection. Histologic sections stained with HE. Infiltrating cells were found in the fat layer between dermis and muscle tissue at sites infected with $\ge 2 \times 10^4$ CFU (A1, B1, C1 and D1). While the two highest inoculation doses led to the development of lesions with a necrotic core surrounded by strong infiltration (A2, B2), infiltration but no necrotic core was observed when doses of 2×10^4 and 2×10^3 CFU were used (C2, D2 insert). Fat cell ghosts were found at sites infected with the three highest inoculation doses (A2, B2 and C2). x: necrosis, y: infiltration, z: fat cell ghosts. doi:10.1371/journal.pntd.0002968.g002

Infection and inoculation with synthetic mycolactone A/B

Specific pathogen-free 2-month-old pigs (Large White) from the in-house breeding unit of the Institute of Virology and Immunology (IVI) were kept under BSL3 conditions one week prior and during the time of experimental infection. Animals were checked once daily for macroscopic signs of infection, had *ad libitum* access to water and were fed daily with complete pelleted food.

Pigs were infected on both flanks at four to six infection sites with 100 µl of *M. ulcerans* suspension, containing 2×10^7 , 2×10^6 , 2×10^5 , 2×10^4 or 2×10^3 CFU. Injection areas were wiped with 70% ethanol and bacterial suspensions injected subcutaneously with a 26G needle. Individual infection sites were encircled with a black marker and the labelling renewed at least once a week. Animals were euthanized at 2.5 weeks or 6.5 weeks post-infection and tissue samples taken as described below.

In addition, the effect of mycolactone was studied directly by injecting 5 μ g or 0.5 μ g of synthetic mycolactone A/B [36] and analysing tissue specimens taken 2.5 weeks later.

Euthanasia and necropsy

Pigs were euthanized by intravenous injection of pentobarbital (150 mg/kg bodyweight) and subsequent exsanguination. Skin tissue at infection sites was extensively excised with a scalpel and scissors, including all layers of the skin down to the fascia, and samples were immediately transferred to 10% neutral-buffered Formalin solution (approx. 4% formaldehyde).

Histopathological analysis

After fixation samples were transferred to 70% ethanol for storage and transport, dehydrated and embedded into paraffin. 5 μ m thin sections were cut, deparaffinised, rehydrated and directly stained with Haematoxylin/Eosin (HE) or Ziehl-Neelsen/Methylene blue (ZN) according to WHO standard protocols [11]. Stained sections were mounted with Eukitt mounting medium (Fluka). Pictures were taken with a Leica DM2500B microscope or with an Aperio scanner.

Results

Macroscopic appearance of *Mycobacterium ulcerans* infected pig skin

In order to assess early effects of the subcutaneous experimental infection of pigs with doses of 2×10^3 to 2×10^7 *M. ulcerans* CFU, injection sites were closely monitored for macroscopic changes of the skin. At 2.5 weeks after injection of the bacteria, first changes in colouration and thickness of the skin became apparent at the sites inoculated with the highest inoculation doses, 2×10^7 and 2×10^6 CFU (Fig. 1, B1). Like nodular BU lesions in humans, these early lesions were elevated, movable, firm and palpable. When these skin areas were excised 2.5 weeks and 6.5 weeks after infection and vertically cut in half after fixation in formalin, roundish yellow structures reflecting coagulative necrosis in the dermis became macroscopically apparent (Fig. 1, B2). A belt with

reddish colour, reflecting infiltrating cells and bleeding into the skin, was observed around the necrotic core. While these structures were larger at sites inoculated with a dose of 2×10^7 CFU than at sites inoculated with 2×10^6 CFU, the general architecture observed with both inoculation doses was similar. At sites inoculated with $< 2 \times 10^5$ CFU, no macroscopically visible alterations of the skin were found 2.5 weeks after infection (Fig. 1, A1 and A2).

At 6.5 weeks after experimental infection, sites injected with the highest inoculation dose had either enlarged to an indurated plaque (Fig. 1 E1) or ulcerated (Fig. 1, D1). At sites injected with 2×10^6 CFU, nodular lesions were observed that were flatter and less palpable compared to those detected 2.5 weeks after infection (Fig. 1, C1). These lesions were macroscopically clearly visible in cross sections through the tissue (Fig. 1, C2). Nodular and ulcerative lesions exhibited greyish/reddish colour changes in the dermis and subcutis (Fig. 1, C2 and D2). The plaque lesion developing after injection with 2×10^7 CFU appeared as long cord-like structure with a centre made of yellowish necrotic slough, surrounded by several layers differing in colouration (Fig. 1, E2).

Histopathological features of the pig skin 2.5 weeks after experimental infection

Microcopically, infiltrating immune cells were found 2.5 weeks after infection at all sites inoculated with $\geq 2 \times 10^4$ CFU (Fig. 2, A1, B1, C1 and D1). As expected from the macroscopically observed signs, the most pronounced histopathological alterations were associated with the two highest inoculation doses (2×10^7 and 2×10^6 CFU). The non-ulcerative lesions that developed between the dermis and the underlying muscle tissue displaced the fat layer (Fig. 2, A1 and B1) and caused the macroscopically visible elevation of the skin (Fig. 1, B1). Microscopically, a necrotic core surrounded by large numbers of infiltrating cells and interspersed with fat cell ghosts was observed (Fig. 2, A2 and B2).

At sites infected with 2×10^5 CFU, no necrotic core structures but some fat cell ghosts and accumulations of infiltrating cells were found (Fig. 2, C1 and C2). The infection with 2×10^4 CFU caused a small accumulation of infiltrating cells (Fig. 2, D1 and D2) and no signs of infection and/or inflammation were observed at sites inoculated with the lowest dose (2×10^3 CFU).

Histopathological features of the pig skin 6.5 weeks after experimental infection

At 6.5 weeks after experimental infection, histopathological changes were only found at sites that had been injected with 2×10^7 or 2×10^6 CFU. In contrast, the skin appeared macro- and microscopically healthy following infection with lower doses of *M. ulcerans*, exhibiting intact epidermis and fat cells, undistorted collagen fibre networks and no marked inflammatory infiltration (Fig. 3, D1 and D2).

Where the infection focus had started to ulcerate, strong infiltration towards the destroyed epidermis was observed (Fig. 3, A1 and A2). No AFB were found in this region, indicative for loss of the necrotic core with the major burden of AFB through the ulceration (Fig. 3, A2, Fig. 4, B1 and B2). Small clusters of AFB

Figure 3. Microscopic appearance of pig skin 6.5 weeks after experimental infection. Histologic sections stained with Haematoxylin/Eosin (HE) (A1, A2, B1, C1, D1 and D2) or Ziehl-Neelsen/Methylene blue (ZN) (B2, C2). At sites injected with 2×10^7 CFU nodules had either developed into a small ulcer with destroyed epidermis (z), strong infiltration and indications for the development of undermined edges (A1, A2, dotted line) or into a plaque with a necrotic core surrounded by infiltrating cells (B1). x: intact epidermis, y: epidermal hyperplasia, z: destroyed/missing epidermis. The site infected with 2×10^6 CFU showed a similar architecture as the plaque but flatter, less organized and with a smaller overall circumference (C1). Both lesions comprised AFB in their necrotic cores, either in big clumps (B2) or in smaller numbers and smaller aggregations (C2). No signs of infection, inflammation and pathology were observed at sites inoculated with 2×10^5 CFU (D1, D2) or less (not shown). doi:10.1371/journal.pntd.0002968.g003

were found at deeper sites in the tissue, lateral to the ulceration site (Fig. 4, B3–B5). Infiltration and destruction of collagen fibres extended into the lower part of the dermis and the upper part of the subcutis, reaching far beyond the area where the epidermis was destroyed (Fig. 3, A1), indicating the formation of undermined edges (Fig. 3, A2, dotted line).

The overall architecture of the plaque lesion that had developed resembled the nodular stages seen 2.5 weeks after infection, i.e. a necrotic centre was surrounded by lavers of infiltrating cells (Fig. 3, B1). While large clumps of extracellular AFB were found in the necrotic core after injection of 2×10^7 CFU (Fig. 3, B2), AFB were less abundant and bacterial clumps smaller when 2×10^6 CFU were used for infection (Fig. 3, C2). Fig. 4A depicts the complex architecture of a plaque lesion $(2 \times 10^7 \text{ CFU dose})$ with several distinct belts of infiltrating cells surrounding a central necrotic core which contained huge clusters of AFB but was completely devoid of infiltration (Fig. 4A, Ring 1, A1 and A2). In the surrounding ring 2, AFB were scarce and had mostly a beaded appearance. In addition to these single AFB, small globi-like clusters of AFB were found, along with Methylene blue stained remains of infiltrating cells (Fig. 4A, Ring 2, A3 and A4). Ring 3 contained mostly small infiltrating cells that appeared intact, and some acid-fast bacterial debris (Fig. 4A, Ring 3, A5 and A6). The outermost layer that could be distinguished did not contain AFB and was mainly built by macrophages and lymphocytes (Fig. 4A, Ring 4, A7). Hence, the number and integrity of AFB decreased from the centre to the periphery of the lesion, whereas the integrity of the cellular infiltration showed an opposite trend, most likely reflecting levels of the cytotoxic macrolide mycolactone decreasing from centre to periphery.

Histopathological resemblance of pig and human BU lesions

All key features of BU pathology in humans were also found in the experimentally infected pig skin. Already 2.5 weeks after infection, coagulative necrosis (Fig. 5, A1), fat cell ghosts (Fig. 5, A2) and extracellular clusters of AFB (Fig. 5, A3 and A4) were detected. Slight epidermal hyperplasia was already observed at 2.5 weeks and became more pronounced 6.5 weeks after infection (Fig. 5, A5–A7). At this time, typical histopathological hallmarks of more advanced human BU lesions also emerged in the infected pig skin, namely formation of granulomas (Fig. 5, A8) and presence of giant cells (Fig. 5, A9). Not only experimental infection with *M. ulcerans* led to these typical alterations in the skin, but also the injection of synthetic mycolactone A/B (Fig. 5B).

Development of satellite infection foci

Besides the general histopathological changes, another similarity to findings in human BU [37] was observed: the formation of satellite infection foci adjacent to the primary lesion. A striking example for this is depicted in Fig. 4B where two satellite foci with small clusters of AFB in a necrotic core were found peripheral to the ulcerated main infection focus (Fig. 4B, Region 2). Likewise in the plaque lesion depicted in Fig. 4A, clusters of AFB were found near the main infection focus (Fig. 4A, Ring 5, A8).

Discussion

Detailed studies on the early pathogenesis of BU in an animal model closely mimicking human BU would be very important for a better understanding of host-pathogen interactions and the relative importance of different effector functions of the innate and adaptive immune system against M. ulcerans. Here we explored the potential of the pig to serve as model for human M. ulcerans infection. After having infected pigs subcutaneously with high doses $(2 \times 10^6 \text{ or } 2 \times 10^7 \text{ CFU})$ of *M. ulcerans* bacteria, we observed the development of different forms of BU lesions (nodules, plaques and ulcers). Macroscopic and histopathological changes closely mirrored human BU. Challenge with lower doses $(2 \times 10^3 \text{ to})$ $2 \times 10^{\circ}$ CFU) resulted in limited tissue destruction and/or infiltration 2.5 weeks after infection, which resolved spontaneously until week 6.5. Likewise, the dose of bacteria transmitted may be of critical importance for the outcome of a natural M. ulcerans infection in humans. Sero-epidemiological analyses in human populations living in BU endemic areas have indicated that exposure to M. ulcerans often leads to self-resolving, nonsymptomatic infections, as indicated by development of M. ulcerans specific antibody responses [38,39]. While macrophages and other immune cells might be able to eliminate smaller numbers of scattered M. ulcerans cells, microcolonies of a critical size may develop a protective cloud of mycolactone around them. If the local concentration of the macrolide cytotoxin exceeds a certain level, infiltrating cells may be killed before they can reach the bacteria. This leads to the characteristic picture of clusters of extracellular AFB located primarily in the necrotic core of advanced lesions, which is devoid of living infiltrating immune cells, but contains debris of early inflammatory infiltrates [40,41].

In our study, we observed round elevations of the skin already 2.5 weeks after infection. These alterations were firm, movable and clearly palpable and hence displayed the characteristic features of human BU nodules [11]. Microscopic investigation of the infected skin sites revealed that most histopathological hallmarks of BU had already developed during the first 2.5 weeks of infection if 2×10^6 or 2×10^7 CFU of *M. ulcerans* was used. The experimentally induced nodules exhibited a necrotic core containing extracellular AFB surrounded by infiltrating cells and fat cell ghosts. Subcutaneous injection of the bacteria led to the formation of an infection focus in the lower dermis and subcutis, where it is also typically found in human BU [13].

In ulcerative human BU lesions AFB are typically focally distributed and not evenly dispersed in the affected tissue [37,42]. Ulceration leads to the shedding of necrotic tissue containing masses of AFB. Therefore the bacterial burden is usually higher in non-ulcerative lesions than in ulcers, where the majority of the remaining AFB reside in the undermined edges of the ulcers. Our histopathological analyses showed, that like in human BU disease [37], satellite lesions may develop near the primary lesion. These may emerge from globi-like accumulations of AFB originating from bacteria that were internalized and transported to distant sites by phagocytic cells. Globi-like accumulations are also found in human BU [43] and in experimentally infected mice [44–46]. Again, these microcolonies may have to reach a critical size to be

570µm

4µm

B5

570µm

Figure 4. Containment of large amounts of AFB in the necrotic core and development of satellite microcolonies. Histologic sections stained with ZN. A plaque (A) and a small ulcer (B) are shown that developed 6.5 weeks after infection with 2×10^7 CFU. The ulcerated lesion was strongly infiltrated at the site of ulceration, where no AFB were found (Region 1, B1, B2). Lateral and between dermis and muscle tissue infiltrating cells enclosed small necrotic areas (Region 2, B3), where AFB were found as satellite microcolonies (B4, B5). The plaque consisted of distinct layers of infiltrating cells encasing a necrotic core containing large clumps of bacteria (Ring 1, A1, A2). A second and third ring with decreasing bacterial load and integrity and increasing integrity of infiltrating cells were layered around this core (Ring 2, A3, A4 and Ring 3, A5, A6). A belt of intact cells was surrounding these three inner layers. It did not contain any AFB (Ring 4, A7) except for a microcolony peripheral to the main bacterial burden (Ring 5, A8).

doi:10.1371/journal.pntd.0002968.g004

able to develop a protective cloud of mycolactone around them. The emergence of only small numbers of newly established microcolonies may explain why borders of advanced ulcers often appear to be very heterogeneous with respect to disease activity with some regions displaying progressive tissue destruction and others showing spontaneous healing tendencies.

At 6.5 weeks after infection with 2×10^6 or 2×10^7 CFU of *M. ulcerans*, lesions that were still closed comprised a necrotic centre

containing clumps of AFB surrounded by well stratified belts of infiltrating cells. Similarly, lesions consisting of a necrotic core surrounded by an inner belt of CD14 positive monocytes/ macrophages and a more external belt of CD3 positive T-cells have been described in human BU [47]. The integrity and number of bacteria was decreasing to the outer rim of the lesion. In contrast, the density and integrity of the cellular infiltrates decreased towards the necrotic core.

(B) Histopathology upon mycolactone injection

Figure 5. Histophathological hallmarks of Buruli ulcer in experimentally infected pig skin. Histologic sections stained with Haematoxylin/ Eosin (HE) (A1, A2, A5, A6, A7, A8, A9, B1, B2 and B3) or Ziehl-Neelsen/Methylene blue (ZN) (A3, A4). A: All typical histopathological features of BU in humans were found in infected pig skin. A1: necrosis, A2: fat cell ghosts, A3 and A4: extracellular clusters of AFB, A5: healthy epidermis, A6: moderate epidermal hyperplasia, A7: strong epidermal hyperplasia, A8: granuloma formation, A9: giant cells. B: Histopathological changes induced by mycolactone injection. B1: necrosis, B2: fat cell ghosts, B3: giant cells. doi:10.1371/journal.pntd.0002968.q005

In the pig model first macroscopic signs of infection (nodules) developed relatively fast after injection of a high number of bacteria. For human BU disease in Uganda and Southern Australia incubation periods of 4–13 weeks and 5–38 weeks have been estimated, respectively [48]. However, incubation periods as short as 2–3 weeks have also been described [49]. Despite extensive analyses we did not find bacteria in the tissue with low inoculation doses at the 6.5 week time point. Therefore we assume that also at later time points lesions would not develop with these low infection doses. It is possible that pigs are more resistant to *M. ulcerans* infection than humans. Consequently, the size of the inoculum to achieve productive experimental infection may be higher for pigs than for the natural infection of humans. This high experimental inoculation dose may have led to fast progression of the disease.

In conclusion, our findings indicate that the pig is a very good animal model to study many aspects of *M. ulcerans* infection. Pig skin represents a much closer model for human skin than murine foot pads, ears or tails with respect to physiology, structure and abundance of fat tissue [50]. In addition, the immune system of the

References

- Buruli ulcer: Mycobacterium ulcerans infection (2000). Geneva: World Health Organization, Global Buruli Ulcer Initiative. 118 p.
- Hayman J (1985) Clinical features of Mycobacterium ulcerans infection. Australas J Dermatol 26: 67–73. doi:10.1111/j.1440-0960.1985.tb01819.x.
- Merritt RW, Walker ED, Small PLC, Wallace JR, Johnson PDR, et al. (2010) Ecology and transmission of Buruli ulcer disease: A systematic review. PLoS Negl Trop Dis 4: e911. doi:10.1371/journal.pntd.0000911.
- Bratschi MW, Bolz M, Minyem JC, Grize L, Wantong FG, et al. (2013) Geographic distribution, age pattern and sites of lesions in a cohort of Buruli ulcer patients from the Mapé Basin of Cameroon. PLoS Negl Trop Dis 7: e2252. doi:10.1371/journal.pntd.0002252.
- WHO (2014) Provisional guidance on the role of specific antibiotics in the management of *Mycobacterium ulcerans* disease (Buruli ulcer). Available: http:// www.who.int/buruli/information/antibiotics/en/index16.html. Accessed 4 February 2014.
- Nienhuis WA, Stienstra Y, Thompson WA, Awuah PC, Abass KM, et al. (2010) Antimicrobial treatment for early, limited *Mycobacterium ulcerans* infection: a randomised controlled trial. The Lancet 375: 664–672. doi:10.1016/S0140-6736(09)61962-0.
- Sarfo FS, Phillips R, Asiedu K, Ampadu E, Bobi N, et al. (2010) Clinical efficacy of combination of rifampin and streptomycin for treatment of *Mycobacterium ulcerans* disease. Antimicrob Agents Chemother 54: 3678–3685. doi:10.1128/ AAC.00299-10.
- Chauty A, Ardant M-F, Adeye A, Euverte H, Guedenon A, et al. (2007) Promising clinical efficacy of Streptomycin-Rifampin combination for treatment of Buruli ulcer (*Mycobacterium ulcerans* disease). Antimicrob Agents Chemother 51: 4029–4035. doi:10.1128/AAC.00175-07.
- Kibadi K, Boelaert M, Fraga AG, Kayinua M, Longatto-Filho A, et al. (2010) Response to treatment in a prospective cohort of patients with large ulcerated lesions suspected to be Buruli ulcer (*Mycobacterium ulcerans* disease). PLoS Negl Trop Dis 4: e736. doi:10.1371/journal.pntd.0000736.
- George KM, Chatterjee D, Gunawardana G, Welty D, Hayman J, et al. (1999) Mycolactone: a polyketide toxin from *Mycobacterium ulcerans* required for virulence. Science 283: 854–857.
- Portaels F, Johnson P, Meyers WM, Initiative WHOGBU (2001) Buruli ulcer: diagnosis of *Mycobacterium ulcerans* disease: a manual for health care providers/ edited by: Françoise Portaels, Paul Johnson, Wayne M. Meyers. Available: http://apps.who.int/iris/handle/10665/67000. Accessed 6 February 2014.
- Guarner J, Bartlett J, Whitney EAS, Raghunathan PL, Stienstra Y, et al. (2003) Histopathologic features of *Mycobacterium ulcerans* infection. Emerg Infect Dis 9: 651–656.
- Hayman J (1993) Out of Africa: observations on the histopathology of Mycobacterium ulcerans infection. J Clin Pathol 46: 5–9.
- Mitchell PJ, McOrist S, Bilney R (1987) Epidemiology of Mycobacterium ulcerans infection in koalas (*Phascolarctos cinereus*) on Raymond Island, southeastern Australia. J Wildl Dis 23: 386–390.
- Mitchell PJ, Jerrett IV, Slee KJ (1984) Skin ulcers caused by Mycobacterium ulcerans in koalas near Bairnsdale, Australia. Pathology (Phila) 16: 256–260.
- Trott KA, Stacy BA, Lifland BD, Diggs HE, Harland RM, et al. (2004) Characterization of a *Mycobacterium ulcerans*-like infection in a colony of African tropical clawed frogs (*Xenopus tropicalis*). Comp Med 54: 309–317.
- O'Brien CR, Handasyde KA, Hibble J, Lavender CJ, Legione AR, et al. (2014) Clinical, Microbiological and pathological findings of Mycobacterium ulcerans

pig resembles the human system more closely than that of the mouse [34]. In particular the development of new therapeutic and prophylactic interventions might benefit from the porcine M. *ulcerans* infection model.

Acknowledgments

We thank Martin Bratschi and Sarah Kerber for providing Cameroonian *M. ulcerans* for this infection experiment. Thank you also to Daniel Brechbühl (Dänu) for taking good care of the pigs before, during and after the experiment. We thank Dr. Masato Murakami, Vincent Romanet, Caroline Stork, Ernesta Dammassa and Patricia Barzaghi Rinaudo from Novartis Basel for excellent technical support and providing access to lab equipment for histopathology. We also thank Peter Schmid for the Aperio scans of the tissue.

Author Contributions

Conceived and designed the experiments: MB MER GZ GP. Performed the experiments: MB NR MTR GZ. Analyzed the data: MB MTR GP. Contributed to the writing of the manuscript: MB MTR GZ GP.

infection in three Australian possum species. PLoS Negl Trop Dis 8: e2666. doi:10.1371/journal.pntd.0002666.

- Elsner L, Wayne J, O'Brien CR, McCowan C, Malik R, et al. (2008) Localised Mycobacterium ulcerans infection in a cat in Australia. J Feline Med Surg 10: 407– 412. doi:10.1016/j.jfms.2008.03.003.
- Portaels F, Chemial K, Elsen P, Johnson PD, Hayman JA, et al. (2001) Mycobacterium ulcerans in wild animals. Rev Sci Tech Int Off Epizoot 20: 252–264.
- Sakaguchi K, Iima H, Hirayama K, Okamoto M, Matsuda K, et al. (2011) Mycobacterium ulcerans infection in an Indian flap-shelled turtle (Lissemys punctata punctata). J Vet Med Sci Jpn Soc Vet Sci 73: 1217–1220.
- Van Zyl A, Daniel J, Wayne J, McCowan C, Malik R, et al. (2010) Mycobacterium ulcerans infections in two horses in south-eastern Australia. Aust Vet J 88: 101– 106. doi:10.1111/j.1751-0813.2009.00544.x.
- Fyfe JAM, Lavender CJ, Handasyde KA, Legione AR, O'Brien CR, et al. (2010) A major role for mammals in the ecology of *Mycobacterium ulcerans*. PLoS Negl Trop Dis 4: e791. doi:10.1371/journal.pntd.0000791.
- Marcus LC, Stottmeier KD, Morrow RH (1976) Experimental alimentary infection of anole lizards (*Anolis carolinensis*) with *Mycobacterium ulcerans*. Am J Trop Med Hyg 25: 630–632.
- Walsh DS, Meyers WM, Krieg RE, Walsh GP (1999) Transmission of Mycobacterium ulcerans to the nine-banded armadillo. Am J Trop Med Hyg 61: 694–697.
- Singh NB, Srivastava A, Verma VK, Kumar A, Gupta SK (1984) Mastomys natalensis: a new animal model for Mycobacterium ulcerans research. Indian J Exp Biol 22: 393–394.
- Addo P, Owusu E, Adu-Addai B, Quartey M, Abbas M, et al. (2005) Findings from a buruli ulcer mouse model study. Ghana Med J 39: 86–93.
- Oliveira MS, Fraga AG, Torrado E, Castro AG, Pereira JP, et al. (2005) Infection with *Mycobacterium ulcerans* induces persistent inflammatory responses in mice. Infect Immun 73: 6299–6310. doi:10.1128/IAI.73.10.6299-6310.2005.
- Read JK, Heggie CM, Meyers WM, Connor DH (1974) Cytotoxic Activity of Mycobacterium ulcerans. Infect Immun 9: 1114–1122.
- Walsh DS, Dela Cruz EC, Abalos RM, Tan EV, Walsh GP, et al. (2007) Clinical and histologic features of skin lesions in a cynomolgus monkey experimentally infected with *Mycobacterium ulcerans* (Buruli ulcer) by intradermal inoculation. Am J Trop Med Hyg 76: 132–134.
- Swindle MM, Makin A, Herron AJ, Clubb FJ Jr, Frazier KS (2012) Swine as models in biomedical research and toxicology testing. Vet Pathol 49: 344–356. doi:10.1177/0300985811402846.
- Meurens F, Summerfield A, Nauwynck H, Saif L, Gerdts V (2012) The pig: a model for human infectious diseases. Trends Microbiol 20: 50–57. doi:10.1016/ j.tim.2011.11.002.
- Sullivan TP, Eaglstein WH, Davis SC, Mertz P (2001) The pig as a model for human wound healing. Wound Repair Regen Off Publ Wound Heal Soc Eur Tissue Repair Soc 9: 66–76.
- Liu Y, Chen J, Shang H, Liu C, Wang Y, et al. (2010) Light microscopic, electron microscopic, and immunohistochemical comparison of Bama minipig (Sus scrofa domestica) and human skin. Comp Med 60: 142–148.
- 34. Dawson H (2011) A comparative assessment of the pig, mouse and human genomes: structural and functional analysis of genes involved in immunity and inflammation. The Minipig in Biomedical Research. CRC Press. 323–342 p. Available: http://www.crcnetbase.com/doi/abs/10.1201/b11356-28. Accessed 4 February 2014.

- Fairbairn L, Kapetanovic R, Sester DP, Hume DA (2011) The mononuclear phagocyte system of the pig as a model for understanding human innate immunity and disease. J Leukoc Biol 89: 855–871. doi:10.1189/jlb.1110607.
- Gersbach P, Jantsch A, Feyen F, Scherr N, Dangy J-P, et al. (2011) A ringclosing metathesis (RCM)-based approach to mycolactones A/B. Chem Weinh Bergstr Ger 17: 13017–13031. doi:10.1002/chem.201101799.
- Rondini S, Horsfield C, Mensah-Quainoo E, Junghanss T, Lucas S, et al. (2006) Contiguous spread of *Mycobacterium ulcerans* in Buruli ulcer lesions analysed by histopathology and real-time PCR quantification of mycobacterial DNA. J Pathol 208: 119–128. doi:10.1002/path.1864.
- Yeboah-Manu D, Röltgen K, Opare W, Asan-Ampah K, Quenin-Fosu K, et al. (2012) Sero-epidemiology as a tool to screen populations for exposure to *Mycobacterium ulcerans*. PLoS Negl Trop Dis 6: e1460. doi:10.1371/journal. pntd.0001460.
- Diaz D, Döbeli H, Yeboah-Manu D, Mensah-Quainoo E, Friedlein A, et al. (2006) Use of the immunodominant 18-kiloDalton small heat shock protein as a serological marker for exposure to *Mycobacterium ulcerans*. Clin Vaccine Immunol CVI 13: 1314–1321. doi:10.1128/CVI.00254-06.
- Ruf M-T, Sopoh GE, Brun LV, Dossou AD, Barogui YT, et al. (2011) Histopathological changes and clinical responses of Buruli ulcer plaque lesions during chemotherapy: a role for surgical removal of necrotic tissue? PLoS Negl Trop Dis 5: e1334. doi:10.1371/journal.pntd.0001334.
- Schütte D, Um-Boock A, Mensah-Quainoo E, Itin P, Schmid P, et al. (2007) Development of highly organized lymphoid structures in Buruli ulcer lesions after treatment with rifampicin and streptomycin. PLoS Negl Trop Dis 1: e2. doi:10.1371/journal.pntd.0000002.
- Peduzzi E, Groeper C, Schütte D, Zajac P, Rondini S, et al. (2007) Local activation of the innate immune system in Buruli ulcer lesions. J Invest Dermatol 127: 638–645. doi:10.1038/sj.jid.5700593.

- Schütte D, Umboock A, Pluschke G (2009) Phagocytosis of Mycobacterium ulcerans in the course of rifampicin and streptomycin chemotherapy in Buruli ulcer lesions. Br J Dermatol 160: 273–283. doi:10.1111/j.1365-2133.2008.08879.x.
- Schütte D, Pluschke G (2009) Immunosuppression and treatment-associated inflammatory response in patients with *Mycobacterium ulcerans* infection (Buruli ulcer). Expert Opin Biol Ther 9: 187–200. doi:10.1517/14712590802631854.
- Ruf M-T, Schütte D, Chauffour A, Jarlier V, Ji B, et al. (2012) Chemotherapyassociated changes of histopathological features of *Mycobacterium ulcerans* lesions in a Buruli ulcer mouse model. Antimicrob Agents Chemother 56: 687–696. doi:10.1128/AAC.05543-11.
- Coutanceau E, Marsollier L, Brosch R, Perret E, Goossens P, et al. (2005) Modulation of the host immune response by a transient intracellular stage of *Mycobacterium ulcerans*: the contribution of endogenous mycolactone toxin. Cell Microbiol 7: 1187–1196. doi:10.1111/j.1462-5822.2005.00546.x.
- 47. Ruf M-T, Chauty A, Adeye A, Ardant M-F, Koussemou H, et al. (2011) Secondary Buruli ulcer skin lesions emerging several months after completion of chemotherapy: paradoxical reaction or evidence for immune protection? PLoS Negl Trop Dis 5: e1252. doi:10.1371/journal.pntd.0001252.
- Trubiano JA, Lavender CJ, Fyfe JAM, Bittmann S, Johnson PDR (2013) The incubation period of Buruli ulcer (*Mycobacterium ulcerans* infection). PLoS Negl Trop Dis 7: e2463. Available: http://www.ncbi.nlm.nih.gov/pmc/articles/ PMC3789762/. Accessed 21 March 2014.
- Reid IS (1967) Mycobacterium ulcerans infection: a report of 13 cases at the Port Moresby General Hospital, Papua. Med J Aust 1: 427–431.
- Addo P, Owusu E, Adu-Addai B, Quartey M, Abbas M, et al. (2005) Findings from a Buruli ulcer mouse model study. Ghana Med J 39: 86–93.